Author's response to reviews

Title: Mycobacterium avium-intracellulare cellulitis occurring with septic arthritis after joint injection: Case report

Authors:

   David M Murdoch (dmurdoch@email.unc.edu)
   Jay R McDonald (jmcdonald@epiresearch-stl.org)

Version: 3 Date: 23 February 2007

Author's response to reviews: see over
February 23, 2007

The BioMed Central Editorial Production Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: corepreaccept@biomedcentral.com
Web: http://www.biomedcentral.com/

Dear BioMed Central Editorial Production Team:

In response to production requests for the manuscript entitled “Mycobacterium avium-intracellulare cellulitis occurring with septic arthritis after joint injection: Case report”, we have made the following changes:

1. Removed the headers from the manuscript.
2. Removed the text 'Title page' from the title page.
3. Removed the word count from the manuscript.
4. Fixed reference #5 to include all authors.
5. Removed underlining from the manuscript.
6. Renamed section to “Competing interests”.
7. Updated the “Acknowledgements” section to include “None”.
8. Added arrow highlighting the rice bodies for Figure 2 in response to reviewer’s request.

We hope these edits are sufficient and look forward to the publication of the manuscript in your journal. Thank you for your time and effort. Please let us know if we can be of any further assistance.

Sincerely,

David M. Murdoch, MD, MPH
CB#7435, McGravran - Greenberg Hall - 2104H
School of Public Health
University of North Carolina at Chapel Hill
Chapel Hill, NC 27599-7435
Email: dmurdoch@email.unc.edu
Phone: 919-843-2726
Fax: 919-966-8928